Author’s response to reviews

Title: Women’s empowerment and fertility preferences in high fertility countries in Sub-Saharan Africa

Authors:

Esso - Hanam ATAKE (atakesyl@gmail.com)
Pitaloumani GNKOU ALI (gnak2005@yahoo.fr)

Version: 5 Date: 14 Mar 2019

Author’s response to reviews:

We are very grateful to the editors for their appropriate and constructive suggestions and for their proposed corrections to improve the paper. We have addressed all the issues raised and have modified the paper accordingly.

1. Title page:
Please list the full names, institutional addresses and email addresses for all authors on the title page. The corresponding author should be indicated.

R: Thank you very much. We take into account your comments.

2. Figure Legend:
Please provide figure legends under a separate heading of 'Figure Legends' after the References. If legends are present within the figure files, please remove them. Figure files should contain only the image, as well as any associated keys/annotations.

R: Please, legends are not present within the figure files. Thank you for your comments.

3. Clean manuscript:

Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

R: Thank you very much. We take into account your comments.